Author's response to reviews

Title: Physical activity, sedentary behaviors, and estimated insulin sensitivity and secretion in pregnant and non-pregnant women

Authors:

Anna Gradmark (anna.gradmark@medicin.umu.se)
Jeremy Pomeroy (pomeroyj@niddk.nih.gov)
Frida Renström (frenstro@hsph.harvard.edu)
Susanne Steiginga (s_steiginga@hotmail.com)
Margareta Persson (margareta.persson@nurs.umu.se)
Antony Wright (Antony.Wright@mrc-epid.cam.ac.uk)
Les Bluck (Les.Bluck@mrc-hnr.cam.ac.uk)
Magnus Domellöf (magnus.domellof@pediatri.umu.se)
Steven E Kahn (skahn@u.washington.edu)
Ingrid Mogren (ingrid.mogren@obgyn.umu.se)
Paul W Franks (paul.franks@med.lu.se)

Version: 4 Date: 4 June 2011

Author's response to reviews:

Thank you for provisionally accepting our paper. We have made all requested changes to the enclosed version of the manuscript.

Best wishes,

Pro. Paul Franks